Author’s response to reviews

Title: Effects of nine weeks isokinetic training on power, golf kinematics, and driver performance in pre-elite golfers.

Authors:
James Parker (james.parker@hh.se;james.parker@scos.se)
Charlie Lagerhem (lagge.lagerhem@hotmail.com)
John Hellström (johnhellstrom@me.com)
M C Olsson (charlotte.olson@hh.se)

Version: 2 Date: 28 Nov 2017

Author’s response to reviews:
Effects of nine weeks isokinetic training on power, golf kinematics, and driver performance in pre-elite golfers.

Dear editor

Here is our submission of the final version of manuscript SSMR-D-17-00031R1. In submission guidelines under the heading 'preparing your manuscript,' we do not find any information about including the heading 'open access'? We have therefore omitted it from our manuscript. If it should be included, we are happy to include it again.

Best regards,
James Parker, Charlie Lagerhem, John Hellström, & M Charlotte Olsson.

Overview of changes
Comments from the editor:
1. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.
Abbreviations, Ethics approval and consent to participate, Consent for publication, Availability of data and material, Competing interests, Funding, Authors' contributions, Acknowledgements

Authors: We have now included the heading ‘Declarations’ with all required subheadings after the ‘Conclusion’ section and before ‘References’, line 469-502.

2. Under the declarations section 'Ethics approval and consent to participate' Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate. Please also indicate whether consent to participate was written or verbal.
Authors: We have now included the full name of the ethics committee, institute and reference number under the subheading ‘Ethics approval and consent to participate’, line 475-477.

3. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.
Authors: The figure 1 heading is now listed under the separate heading ‘Figure Legends’ after ‘References’, line 623-627.

4. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated
Authors: Email addresses for all authors are now included on the title page and the corresponding author is indicated with an asterisk (*).

5. Your manuscript should be uploaded as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Authors: The manuscript and figure 1 in the article are uploaded as a separate file.